Author's response to reviews

Title: Hospital-acquired invasive pulmonary aspergillosis in patients with hepatic failure

Authors:

Dan Li (lidan@shaphc.org)
Liang Chen (chenliang@shaphc.org)
Xian Ding (dingxian@shaphc.org)
Ran Tao (taoran@shaphc.org)
YongXin Zhang (mzld@hotmail.com)
JieFei Wang (immu_lab@hotmail.com)

Version: 6 Date: 12 June 2008

Author's response to reviews:

We correct formatting of text and tables according to the journal instructions as requested by the editor, as well as the language corrections.

The details list below:

Page 2, Line 2: Background: Invasive pulmonary aspergillosis (IPA) is a rapid progressive, fatal disease that occurs mostly in immunocompromised patients With patients with severe liver disease being at a heightened risk for infections.
It was corrected into: Invasive pulmonary aspergillosis (IPA) is a rapid, progressive, fatal disease that occurs mostly in immunocompromised patients. Patients with severe liver disease are at a heightened risk for infections.

Page 3, Line 21: The medical and microbiological records as well as the chest radiographic imaging of all patients with hepatic failure and accompanying lung infection were reviewed.
It was corrected into: Medical and microbiological records as well as chest radiographic imaging of all patients with hepatic failure and accompanying lung infection were reviewed.

Page 4, Line 1: Probable IPA was defined as the demonstration of filamentous fungi compatible with the morphology of Aspergillus and/or a positive culture for Aspergillus from bronchoalveolar lavage (BAL) specimen.
It was corrected into: Probable IPA was defined as the demonstration of filamentous fungi compatible with the morphology of Aspergillus and/or a positive culture for Aspergillus from bronchoalveolar lavage (BAL) specimen.

Page 4, Line 15: Single patient hospital room was provided for the patient with high index of suspicion of IPA.
It was corrected into: Single-bed hospital room was provided for the patient with high index of suspicion of IPA.
Page 5, Line 21: Among the biopsied patients …
It was corrected into: Among the patients who underwent biopsy …

Page 8: Author’s contributions: Jie Fei Wang and Yong Xin Zhang designed the study, the other authors collected and analyzed the data; Jie Fei Wang, Dan Li and Liang Chen drafted the manuscript.
It was corrected into: Author’s contributions:
All authors contributed to the design of the study.
Acquisition of data and quality control: Liang Chen, Xian Ding, Ran Tao
Analysis and interpretation of data: Dan Li, Yong Xin Zhang, Jie Fei Wang
Draft of manuscript: Dan Li
Laboratory assessment: Liang Chen, Jie Fei Wang
All authors have read and approved the final manuscript

Page 9: All of the references were corrected according to the journal instructions.